Author's response to reviews

Title: How well can physicians manage Tuberculosis? A public-private sector comparison from Karachi, Pakistan.

Authors:

Maliha Naseer (maliha.naseer@aku.edu)
Ali Khawaja (alianwar90@hotmail.com)
Amin Pethani (amin.pethani@aku.edu)
Salik Aleem (salikaleem@hotmail.com)

Version: 4 Date: 2 October 2013

Author's response to reviews:

Dear Editor,

We are extremely grateful to the reviewers for their constructive feedback on our manuscript.

All the changes made in the manuscript are highlighted RED and BOLD

Point by point reply to reviewers’ comments is as follows:

Reviewer 1:

RESULT SECTION

In last paragraph change “However, treating #10 TB patients in the last one year was found to be significantly associated with [OR 1.57; 95% CI: 1.258-1.996] adequate knowledge scores.” to “However, treating #10 TB patients in the last one year was found to be significantly associated [OR 1.57; 95% CI: 1.258-1.996] with adequate knowledge scores.”

Answer: Suggested changes are incorporated

TABLES 1, 2 AND 3

In the comment under the tables change *p-value >0.05 were considered as significant to p-value <0.05 were considered as significant

Answer: Suggested changes are incorporated

Reviewer 2:

No further revisions were suggested. We thank the reviewer for his constructive feedback.

With best regards,

Maliha Naseer
Email: maliha.naseer@gmail.com.